Author's response to reviews

Title: Higher prevalence and characterization of hepatitis C virus/hepatitis B virus infection in an epidemic area of sharing needle in North China

Authors:

Cheng-Jun Xu (xcj123321@163.com)
Cui-Ping Zhang (zhcp.1979@163.com)
Bi-Fen Luo (luobifen@qq.com)
Li-Jun Liu (986224066@qq.com)
Yun-Zhong Wang (wyz18831419811@163.com)
Xiao-Hong Wang (1214738844@qq.com)
Qiu-Jie He (heqiujie666@163.com)
Shan-Shan Zhou (639818719@qq.com)
Wei-Shan Guo (xyy6632526@163.com)
Jiu-Heng Wang (281630718@qq.com)
Rui-Feng Yang (yangruifeng103@live.cn)
Hai-Ying Zhang (zanghaiying_zhy@163.com)
Hui-Ying Rao (rao.huiying@163.com)
Bo Feng (xyfyfb_1@sina.com)
Lai Wei (weilai@pkuph.edu.cn)

Version: 2
Date: 12 August 2014

Author's response to reviews:

We have included line number, page number, and email addresses for all of the authors on the title page of our manuscript. Page breaks were deleted. The previous cover letter has not been changed.